Assessment of Transmission in Trachoma Programs over Time Suggests No Short-Term Loss of Immunity

Fengchen Liu¹, Travis C. Porco^{1,2,3}*, Kathryn J. Ray¹, Robin L. Bailey⁴, Harran Mkocha⁵, Beatriz Muñoz⁶, Thomas C. Quinn⁷, Thomas M. Lietman^{1,2,3}, Sheila K. West⁶

1 F.I. Proctor Foundation, University of California, San Francisco, California, United States of America, 2 Department of Ophthalmology, University of California, San Francisco, California, United States of America, 3 Department of Epidemiology and Biostatistics, University of California, San Francisco, California, United States of America, 3 Department of Epidemiology and Biostatistics, University of California, San Francisco, California, United States of America, 4 Faculty of Infectious and Tropical Diseases, Clinical Research Department, London School of Hygiene and Tropical Medicine, London, United Kingdom, 5 Kongwa Trachoma Project, Kongwa, Tanzania, 6 Wilmer Eye Institute, Johns Hopkins Hospital, Baltimore, Maryland, United States of America, 7 Johns Hopkins Center for Global Health, Johns Hopkins University, Baltimore, Maryland, United States of America

Abstract

Trachoma programs have dramatically reduced the prevalence of the ocular chlamydia that cause the disease. Some have hypothesized that immunity to the infection may be reduced because of program success in reducing the incidence of infection, and transmission may then increase. Longitudinal studies of multiple communities would be necessary to test this hypothesis. Here, we quantify transmission using an estimated basic reproduction number based on 32 communities during the first, second, and third years of an antibiotic treatment program. We found that there is little to no increase in the basic reproduction number over time. The estimated linear trend in the basic reproduction number, \hat{r}_0 , was found to be -0.025 per year, 95% CI -0.167 to 0.117 per year. We are unable to find evidence supporting any loss of immunity over the course of a 3-year program. This is encouraging, as it allows the possibility that repeated mass antibiotic distributions may eliminate infection from even the most severely affected areas.

Citation: Liu F, Porco TC, Ray KJ, Bailey RL, Mkocha H, et al. (2013) Assessment of Transmission in Trachoma Programs over Time Suggests No Short-Term Loss of Immunity. PLoS Negl Trop Dis 7(7): e2303. doi:10.1371/journal.pntd.0002303

Editor: Joseph M. Vinetz, University of California San Diego School of Medicine, United States of America

Received December 15, 2012; Accepted May 18, 2013; Published July 11, 2013

This is an open-access article, free of all copyright, and may be freely reproduced, distributed, transmitted, modified, built upon, or otherwise used by anyone for any lawful purpose. The work is made available under the Creative Commons CC0 public domain dedication.

Funding: The Bill and Melinda Gates Foundation (grant number 48027; www.gatesfoundation.org) was the main supporter of this trial. This project was also supported by the National Institutes of Health (NIH/NEI K23 EYO19881-01; www.nih.gov), a Models of Infectious Disease Agent Study (MIDAS; www.epimodels. org) grant from the National Institute of General Medical Sciences (http://www.nigms.nih.gov/) at the National Institutes of Health (RFA-GM-11-002) to the University of California, San Francisco (U01-GM087728), and Research to Prevent Blindness. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

* E-mail: travis.porco@ucsf.edu

Introduction

The World Health Organization has targeted trachoma for elimination by the year 2020 [1]. Repeated mass oral azithromycin distributions have been a cornerstone of the treatment strategy. Theoretically, repeated treatments may eventually eliminate infection from even the most severely affected areas [2,3]. In practice, distributions have dramatically reduced the prevalence of infection in a number of locations [3,4,5,6,7,8,9]. However, there remains concern that resistance may develop or that loss of immunity may prevent complete elimination [10,11,12,13,14]. While no stable chlamydial drug resistance has yet been observed, a loss of immunity is possible. Individuals who had not been exposed to infection recently might have less protection than they had had when the infection was more prevalent. An increase in transmission during the course of a program could indicate loss of immunity. Multiple communities would need to be monitored, to assess whether random fluctuations may explain observed differences over time. Here, we analyzed multiple communities from the Tanzanian portion of the Program for the Rapid Elimination of Trachoma trial (PRET [15]) using a stochastic model of transmission, to assess the initial reproductive number for transmission over time.

Methods

Clinical and laboratory results

Communities were monitored as part of a cluster-randomized, trachoma treatment trial in Tanzania [15,16]. In brief, 32 communities in Tanzania were randomized in a two by two factorial design. The first factor was the use of standard versus enhanced coverage with annual mass antibiotic treatment; the second factor was the use or disuse of a rule whereby mass antibiotic administration would be discontinued based on ongoing monitoring. In fact, the use of this rule never led to discontinuation of mass antibiotic administration during the first three years. Thus, all 32 communities received treatment at baseline, 12, and 24 months. A baseline census was conducted in all 32 communities, and again at 12, 24, and 36 months. One hundred randomly selected children aged 0-5 years were examined at baseline, and at 6, 12, 18, 24, 30, and 36 months post baseline. A dacron swab was passed 3 times over their inverted right upper conjunctiva, and processed for the presence of chlamydial DNA as previously described [15]. Our stochastic transmission model was fit to the estimated prevalence of infection at 6, 12, 18, 24, 30, and 36 months.

Author Summary

Trachoma, caused by repeated infections by the ocular strains of Chlamydia trachomatis, is the most common infectious cause of blindness in the world. Treatment for trachoma includes mass azithromycin treatments to the entire community. To reduce the prevalence of infection, the World Health Organization (WHO) advocates at least three annual community-wide distributions of oral antibiotics in affected areas, with further mass treatments based on the prevalence of trachoma. Trachoma programs have dramatically reduced the community prevalence of infection, and some have argued that lowered prevalence of infection may lead to reductions in immunity, and that less immunity may in turn lead to increased transmission from what infection remains. Here, we used a stochastic transmission model to analyze data collected from a 3year antibiotic treatment program (a 32-community, cluster-randomized clinical trial in Tanzania) to assess whether or not transmission actually increases during elimination campaigns. We found no evidence supporting any increase in transmission over the course of the program. The absence of a short term increase in transmission as the prevalence decreases is good news for trachoma programs.

Ethics statement

The study received ethical approval from institutional review board (IRB) of the Johns Hopkins University School of Medicine, the University of California San Francisco, and the Tanzanian National Institute for Medical Research, and was carried out in accordance with the Declaration of Helsinki. All subjects provided informed consent. The informed consent given was oral, because 1) verbal consent is the most ethical way to obtain consent, due to the high illiteracy rates in the study area, 2) IRB approved the use of the oral consent procedure for this study, 3) this oral consent is documented on the registration form for each study participant prior to examination in the field.

Modeling methods

We constructed a stochastic transmission model of transmission of Chlamydia trachomatis infection over time [17,18,19,20]. For community j (j = 1, ..., 32), we assumed a population of size N_j , taken from the number of pre-school children found in the census at the time of treatment (baseline, 12 months, or 24 months). We assumed a classical SIS (susceptible-infective-susceptible) model structure, assuming that the force of infection is proportional to the prevalence of infection in the population with proportionality constant β , and a constant per-capita recovery rate γ (month⁻ Between periods of treatment, we assumed that the probability $p_i(t)$ that there are i infectives in the population obeyed the following equations:

$$\frac{dp_0}{dt} = -\gamma p_1,$$

$$\frac{dp_i}{dt} = \beta \frac{(i-1)(N-i+1)}{N} p_{i-1} + \gamma(i+1)p_{i+1}$$

$$-\beta \frac{i(N-i)}{N} p_i - \gamma i p_i \text{ for } 1 \le i \le N-1,$$

$$(1)$$

and

$$\frac{dp_N}{dt} = \beta \frac{N-1}{N} p_{N-1} - \gamma N p_N.$$

These equations were applied to each village j, though we have suppressed a subscript j for clarity in Equation (1).

Statistical methods

To estimate the transmission coefficient, we used data collected six months and twelve months after each treatment. The model was fit to each of three years. For comparing transmission rates, we initialized the model with observations taken six months after treatment, and we estimated the transmission parameter based on values observed six months after that. Thus, we modeled the time periods from 6 to 12 months, 18 to 24 months, and 30 to 36 months.

For each year, initial values for $p_i(t)$ were determined from as follows. From a population of size \mathcal{N}_i of which the number \mathcal{Y} of infectives equals *i*, the probability P(S=s|Y=i) that *s* positives are observed from a sample of size M_i are sampled is given by the

hypergeometric distribution: $\frac{\begin{pmatrix} i\\ s \end{pmatrix} \begin{pmatrix} N_j - i\\ M_j - s \end{pmatrix}}{\begin{pmatrix} N_j\\ M_j \end{pmatrix}}$. We assumed a beta-binomial prior $P(Y = y) = \begin{pmatrix} N_j\\ y \end{pmatrix} \frac{B(y + \mu, N_j - y + \rho)}{B(\mu, \rho)}$ (where

the shape parameters μ and ρ were computed from the observed distribution of infection of 32 villages at 6 months, 18 months and 30 months, and B(x,y) is the beta function [21])) for all values of y. Application of Bayes' theorem yields

$$p_{i} = P(Y = i | S = s) = \frac{P(S = s | Y = i)}{\sum_{i=0}^{N_{j}} P(S = s | Y = i)}.$$
(2)

For each community j, we used the most recent census data to determine the community size N_i . The initial condition was determined from Equation (2), and the system numerically integrated for six months. Let S_i be the number of positive individuals detected in the sample at the end of the period (for community j). Given the number i of infected individuals, we computed the probability of the observed data according to (:) (N)

$$P(S_j = s) = \sum_{i=0}^{N_j} p_i \frac{\binom{l}{s}\binom{N_j - l}{M_j - s}}{\binom{N_j}{M_i}} \text{ (where } M_j \text{ here denotes the}$$

sample size at the end of the period). We assumed independent communities, and thus we might maximize the sum of the logarithm of the above expressions (summing over all communities).

We assumed specific values of γ (see Table 1) and estimated the value of β that maximizes the total loglikelihood given γ . Standard errors were obtained from the observed Fisher information. We estimated the change, m (year⁻¹), in the transmission coefficient per year by finding maximum likelihood estimates of β_1 and m (with $\beta_2 = \beta_1 + m$, $\beta_3 = \beta_1 + 2m$). For statistical comparison, we also used all time periods to estimate a single (constant) transmission coefficient for all three periods. The basic reproduction number is given by $R_0 = \beta / \gamma$; thus, estimated values of \hat{R}_0 may be computed by dividing the estimated transmission coefficient by γ . The estimated annual change in the basic reproduction number, \hat{r}_0 , may be computed by \hat{m}/γ .

Table 1. Estimated initial reproductive number and the change per year.

| Scenario | Mean duration of infection (1/次) | Estimated \hat{R}_0 (95% CI) | | | | \hat{r}_0 (95% CI) * |
|-------------------------|----------------------------------|--------------------------------|----------------------|----------------------|----------------------|-----------------------------|
| | | Overall | 1 st year | 2 nd year | 3 rd year | |
| Base Case | 6-month | 1.39 (1.28, 1.49) | 1.40 (1.26, 1.55) | 1.38 (1.09, 1.67) | 1.35 (0.92, 1.78) | -0.025 (-0.167, 0.117) |
| Sensitivity Analyses | 18-month | 1.90 (1.62, 2.19) | 11.87 (1.48, 2.26) | 1.92 (1.15, 2.70) | 1.96 (0.82, 3.13) | 0.054 (-0.326, 0.434) |
| | 12-month | 1.64 (1.45, 1.84) | 1.64 (1.37, 1.91) | 1.65 (1.12, 2.18) | 1.66 (0.87, 2.45) | 0.012 (-0.248, 0.273) |
| | 3-month | 1.26 (1.19, 1.31) | 1.28 (1.20, 1.37) | 1.24 (1.08, 1.41) | 1.20 (0.96, 1.45) | -0.041 (-0.122, 0.039) |

*: the 95% CI of \hat{r}_0 was obtained from the observed Fisher information.

doi:10.1371/journal.pntd.0002303.t001

We also estimated an alternative model in which instead of varying the transmission coefficient over time, we instead assumed a constant transmission coefficient, and instead modeled the recovery rate in year i (i=1,2,3) according to $\gamma_i = \gamma_1 + m'(i-1)$, where m' (month⁻¹ year⁻¹) is the annual change in recovery rate.

Previous models have estimated the duration of infection $1/\gamma$ to be from 3 months to 12 months [17,19,20]. As a base case, we assumed a mean duration $1/\gamma$ of 6 months; as a sensitivity analysis, we varied the mean duration from 3 to 18 months.

All calculations were performed using R (version 2.14.1, R Foundation for Statistical Computing, Vienna, Austria).

Results

The numbers of 0–5 year-old children tested for the presence of ocular chlamydia were 3199 (baseline), 3198 (month 6), 3191 (month 12), 3200 (month 18), 3199 (month 24), 3194 (month 30), and 3153 (month 36). The estimated prevalence of ocular chlamydial infection by PCR at baseline was 22.0% (standard deviation 10.1%), at 6 months 10.5% (SD 4.7%), at 12 months 13.0% (SD 6.4%), at 18 months 7.1% (SD 4.4%), at 24 months 8.6% (SD 7.3%), at 30 months 3.5% (SD 2.5%), and at 36 months 4.7% (SD 3.3%).

Assuming a mean duration of infection of six months with betabinomial prior (choosing the shape parameters to match the observed mean and variance of all villages cross sectionally at each initialization time), we found the basic reproduction number to be $\hat{R}_0 = 1.39$ (95% CI: 1.28 to 1.49). For the first year, $\hat{R}_0 = 1.40$ (95% CI: 1.26 to 1.55); for the second year, $\hat{R}_0 = 1.38$ (95% CI: 1.09 to 1.67), and for the third year, $\hat{R}_0 = 1.35$ (95% CI: 0.92 to 1.78).

The estimated change per year in the reproductive number, \hat{r}_0 , was found to be -0.025, 95% CI -0.167 to 0.117, see Table 1. Similar findings were obtained when we assumed other values for the mean duration of infection: three months, $\hat{r}_0 = -0.041$ (95% CI: -0.122 to 0.039) year⁻¹; twelve months, $\hat{r}_0 = 0.012$ (95% CI: -0.248 to 0.273) year⁻¹; eighteen months, $\hat{r}_0 = 0.054$ (95% CI: -0.326 to 0.434) year⁻¹. Regardless of the assumed duration of infection, we find point estimates for the annual change in the basic reproduction number which are near zero. The confidence intervals are wider when a longer duration of infection is assumed, and these intervals include zero (i.e., no change) in every scenario.

Similar findings were obtained when a different choice of prior was used. Specifically, we assumed a uniform distribution as the prior distribution for the number of infected individuals; this yielded an overall $\hat{\mathbf{R}}_0 = 1.30$ (95% CI: 1.19 to 1.41) based on a pooled estimate assuming a constant β for all three years (i.e. assuming m = 0) and a mean duration of infection of six months.

The corresponding estimate for the estimated change per year in the reproductive number is $\hat{r}_0 = -0.084$ (95% CI: -0.227 to 0.058) year⁻¹. Choosing other values of the mean duration together with the uniform prior similarly yielded the following results: three months, $\hat{r}_0 = -0.072$ (95% CI: -0.152 to 0.009) year⁻¹; twelve months, $\hat{r}_0 = -0.103$ (95% CI: -0.366 to 0.161) year⁻¹; eighteen months, $\hat{r}_0 = -0.118$ (95% CI: -0.502 to 0.266) year⁻¹.

We estimated the change in the recovery rate γ , assuming a constant transmission coefficient (the optimal value when infection duration was assumed to be 6 months). The estimated linear trend in recovery rate was found to be 0.013 (95% CI: -0.009 to 0.035) month⁻¹ year⁻¹, with the estimated recovery rate in the first year given by 0.177 (95% CI: 0.154 to 0.20) month⁻¹ year⁻¹. This model yields a substantially similar interpretation as the previous model.

Discussion

Using a transmission model and data collected from a 32community, cluster-randomized clinical trial in Tanzania, we found no evidence of increased transmission from the 1st through the 3rd year of treatment. In fact, our estimates of the reproduction number of the infection were very similar for each year, suggesting no loss of immunity.

Others have proposed an arrested immunity hypothesis, in which the development of protective immune responses is decreased as the duration of chlamydial infection is decreased. It has been suggested that an increased incidence of infection in the presence of a decreased seroprevalence in Finland and British Columbia is due to this phenomenon [10,22]. Trachoma programs have offered an ideal setting to test this hypothesis. A study in Vietnam suggested that a single community with more antibiotic treatment had more rapid return of infection than another less treated community, and that this may be due to a loss of immunity [11]. Without a larger number of treated communities, it was not possible to assess whether the magnitude of this paradoxical result would have been expected by chance alone [23]. In this present study, the large number of longitudinally monitored communities offered more power, yet we were unable to document any evidence of increased transmission with more treatment.

There are several reasons this analysis of these data might fail to detect an increase in transmission, even if such an increase in fact exists. Two years may not be long enough for immunity to wane, although the period in which an earlier study suggested waning was only one year [11]. Uncertainty in the average duration of infection is a potential source of model misspecification, although a sensitivity analysis suggests that the reproductive number and estimated change are not sensitive to the value of infection duration. Even though the trial from which these data came is one of the larger trachoma studies performed, 32 communities may not be a large enough sample size to detect a modest increase in transmission. It is possible that a loss of immunity did occur, but that any effect on transmission was balanced by a decrease in transmission due to other factors; other studies have reported that per-infectious case transmission may decrease with decreasing prevalence, perhaps due to a decrease in the diversity of strains at lower prevalence [19,24]. Finally, we have assumed that transmission is proportional to the number of infectious cases and number of susceptible cases (mass action); if this is not the case, then this may have masked increased transmission at the later, lower prevalence [19].

Models have predicted that if transmission per infectious case remains constant, repeated distributions can eliminate infection from even the most severely affected communities [2,3]. Longitudinal studies have confirmed that local elimination is possible [5,6,25,26,27]. However, these successes might not hold in the

References

- 1. Mariotti SP (2004) New steps toward eliminating blinding trachoma. N Engl J Med 351: 2004–2007.
- Lietman T, Porco T, Dawson C, Blower S (1999) Global elimination of trachoma: how frequently should we administer mass chemotherapy? Nature Medicine 5: 572–576.
- Melese M, Chidambaram JD, Alemayehu W, Lee DC, Yi EH, et al. (2004) Feasibility of eliminating ocular Chlamydia trachomatis with repeat mass antibiotic treatments. JAMA 292: 721–725.
- Schachter J, West ŠK, Mabey D, Dawson CR, Bobo L, et al. (1999) Azithromycin in control of trachoma. Lancet 354: 630–635.
- Gaynor BD, Miao Y, Cevallos V, Jha H, Chaudary JS, et al. (2003) Eliminating trachoma in areas with limited disease. Emerg Infect Dis 9: 596–598.
- Solomon AW, Holland MJ, Alexander ND, Massae PA, Aguirre A, et al. (2004) Mass treatment with single-dose azithromycin for trachoma. N Engl J Med 351: 1962–1971.
- West SK, Munoz B, Mkocha H, Holland MJ, Aguirre A, et al. (2005) Infection with Chlamydia trachomatis after mass treatment of a trachoma hyperendemic community in Tanzania: a longitudinal study. Lancet 366: 1296–1300.
- Chidambaram JD, Alemayehu W, Melese M, Lakew T, Yi E, et al. (2006) Effect of a single mass antibiotic distribution on the prevalence of infectious trachoma. JAMA 295: 1142–1146.
- House JI, Ayele B, Porco TC, Zhou Z, Hong KC, et al. (2009) Assessment of herd protection against trachoma due to repeated mass antibiotic distributions: a cluster-randomised trial. Lancet 373: 1111–1118.
- Brunham RC, Pourbohloul B, Mak S, White R, Rekart ML (2005) The unexpected impact of a Chlamydia trachomatis infection control program on susceptibility to reinfection. J Infect Dis 192: 1836–1844.
- Atik B, Thanh TT, Luong VQ, Lagree S, Dean D (2006) Impact of annual targeted treatment on infectious trachoma and susceptibility to reinfection. Jama 296: 1488–1497.
- Solomon AW, Mohammed Z, Massae PA, Shao JF, Foster A, et al. (2005) Impact of mass distribution of azithromycin on the antibiotic susceptibilities of ocular Chlamydia trachomatis. Antimicrob Agents Chemother 49: 4804–4806.
- Hong KC, Schachter J, Moncada J, Zhou Z, House J, et al. (2009) Lack of macrolide resistance in Chlamydia trachomatis after mass azithromycin distributions for trachoma. Emerg Infect Dis 15: 1088–1090.
- Mabey D, Bailey R, Solomon A, Burton M, Gilbert C, et al. (2007) Targeted treatment of active trachoma. JAMA 297: 588; author reply 589–590.
- Stare D, Harding-Esch E, Munoz B, Bailey R, Mabey D, et al. (2011) Design and baseline data of a randomized trial to evaluate coverage and frequency of

future, if antibiotic resistance were to develop, or if a loss of immunity resulted in increased transmission. The absence of a short term increase in transmission as the prevalence decreases is good news for trachoma programs.

Acknowledgments

We thank the data and safety monitoring committee including Douglas Jabs, MD, MBA (chair), Antoinette Darville, MD, Maureen Maguire, PhD, and Grace Saguti, MD, who were generous with their time and advice, and met before and during the study.

Author Contributions

Conceived and designed the experiments: TML TCP FL SKW. Performed the experiments: FL TCP TML SKW HM BM. Analyzed the data: FL TCP TML BM KJR RLB TCQ. Contributed reagents/materials/analysis tools: SKW TML TCP FL. Wrote the paper: TCP FL TML KJR SKW RLB. Contributed to the collection of clinical data: SKW BM HM. Contributed to the cleaning and management of data: BM.

mass treatment with azithromycin: the Partnership for Rapid Elimination of Trachoma (PRET) in Tanzania and The Gambia. Ophthalmic epidemiology 18: 20–29.

- Harding-Esch EM, Edwards T, Mkocha H, Munoz B, Holland MJ, et al. (2010) Trachoma prevalence and associated risk factors in the gambia and Tanzania: baseline results of a cluster randomised controlled trial. PLoS Negl Trop Dis 4: e861.
- Ray KJ, Lietman TM, Porco TC, Keenan JD, Bailey RL, et al. (2009) When can antibiotic treatments for trachoma be discontinued? Graduating communities in three African countries. PLoS Negl Trop Dis 3: e458.
- Blake IM, Burton MJ, Bailey RL, Solomon AW, West S, et al. (2009) Estimating Household and Community Transmission of Ocular Chlamydia trachomatis. PLoS Negl Trop Dis 3: e401.
- Lietman TM, Gebre T, Ayele B, Ray KJ, Maher MC, et al. (2011) The epidemiological dynamics of infectious trachoma may facilitate elimination. Epidemics 3: 119–124.
- Ray KJ, Porco TC, Hong KC, Lee DC, Alemayehu W, et al. (2007) A rationale for continuing mass antibiotic distributions for trachoma. BMC Infect Dis 7: 91.
- Abramowitz M, Stegun IA, editors (1972) Handbook of mathematical functions. Washington DC: National Bureau of Standards, AMS 55, US Government Printing office.
- Brunham RC, Rekart ML (2008) The arrested immunity hypothesis and the epidemiology of chlamydia control. Sexually transmitted diseases 35: 53–54.
- Mabey D, Bailey R, Solomon A, Burton M, Gilbert C, et al. (2007) Targeted treatment of active trachoma. JAMA : the journal of the American Medical Association 297: 588; author reply 589–590.
- Zhang J, Lietman T, Olinger L, Miao Y, Stephens RS (2004) Genetic diversity of Chlamydia trachomatis and the prevalence of trachoma. Pediatr Infect Dis J 23: 217–220.
- Gill DA, Lakew T, Alemaychu W, Melese M, Zhou Z, et al. (2008) Complete Elimination Is a Difficult Goal for Trachoma Programs in Severely Affected Communities. Clin Infect Dis 46(4):564–6.
- Biebesheimer JB, House J, Hong KC, Lakew T, Alemayehu W, et al. (2009) Complete Local Elimination of Infectious Trachoma from Severely Affected Communities after Six Biannual Mass Azithromycin Distributions. Ophthalmology 116(11):2047–50.
- Melese M, Alemayehu W, Lakew T, Yi E, House J, et al. (2008) Comparison of annual and biannual mass antibiotic administration for elimination of infectious trachoma. JAMA 299: 778–784.